Author's response to reviews

Title: A case of masked toxic adenoma in a patient with non-thyroidal illness

Authors:

Eun Ae Cho (cea202@hanmail.net)
Min Gui Han (neothis7@hanmail.net)
Hyoung Ju Hong (h2jforever@naver.com)
Jee Hee Yoon (babogarden@naver.com)
Jung Sik Moon (doctor-mjs@hanmail.net)
Doo-Young Lee (aqua220@hanmail.net)
Hee Kyung Kim (albeppy@gmail.com)
Ho-Cheol Kang (drkang@chonnam.ac.kr)

Version: 2 Date: 25 April 2013

Author's response to reviews:

Dear Editor-in-Chief of BMC Endocrine Disorders

Here, we submit a revised manuscript entitled “A case of masked toxic adenoma in a patient with non-thyroidal illness syndrome” for publication in BMC Endocrine Disorders.

We added the consent, conflicts of interest, authors’ contributions and acknowledgement in the revised manuscript.

Thank you for your kind consideration and we are looking forward to hearing good news.

With Best Regards

Hee Kyung Kim, MD
Department of Internal Medicine, Chonnam National University Hwasun Hospital, Chonnam National University Medical School
322 Seoyang-ro, Hwasun-eup, Hwasun-gun, Jeonnam, 519-763, South Korea
Tel: +82-61-379-7621
Fax: +82-61-379-7628
E-mail: albeppy@gmail.com